Author’s response to reviews

Title: Screening with the double surprise question to predict deterioration and death: an explorative study

Authors:

Carel Veldhoven (carel.veldhoven@radboudumc.nl)
Nienke Nutma (nienke.nutma@radboudumc.nl)
Willemijn de Graaf (wdegraaf@huisartsenpraktijkbergendal.nl)
Henk Schers (henk.schers@radboudumc.nl)
Stans Verhagen (stans.verhagen@radboudumc.nl)
Kris Vissers (kris.vissers@radboudumc.nl)
Yvonne Engels (yvonne.engels@radboudumc.nl)

Version: 2 Date: 29 Oct 2019

Author’s response to reviews:

Dear professor Cohen,

Thank you very much for your compliments about our study, and the thorough reading of the adapted version.

We changed one sentence on page 6, lines 136-137 (in red). We really appreciate this comment, as we consider it very relevant that the procedure is clearly described in the manuscript.

Thank you for considering our manuscript for publication,

Best regards, also on behalf of the other authors,

Carel Veldhoven